Author's response to reviews

Title: Four cycles of paclitaxel and carboplatin as adjuvant treatment in early-stage ovarian cancer: the experience of the Hellenic Cooperative Oncology Group

Authors:

Aristotle Bamias (abamias@med.uoa.gr)
Christos Papadimitriou (abamias@med.uoa.gr)
Eleni Efstathiou (abamias@med.uoa.gr)
Alexandros Rodolakis (abamias@med.uoa.gr)
Georgios Vlahos (abamias@med.uoa.gr)
Zannis Voulgaris (abamias@med.uoa.gr)
Georgios Bozas (gbozas@med.uoa.gr)
Georgios Fountzilas (fountzil@med.auth.gr)
Gerassimos Aravantinos (hecogaga@otenet.gr)
Evagelia Razis (edrazis@hol.gr)
Dimitra Gika (mdimop@med.uoa.gr)
Meletios A. Dimopoulos (mdimop@med.uoa.gr)

Version: 5 Date: 22 September 2006

Author’s response to reviews:

Alexander Pemberton, PhD
BMC Cancer Athens 21 September 2006

Re: MS 1103156207983361
FOUR CYCLES OF PACLITAXEL AND CARBOPLATIN AS ADJUVANT TREATMENT IN EARLY-STAGE OVARIAN CANCER: A SIX-YEAR EXPERIENCE OF THE HELLENIC COOPERATIVE ONCOLOGY GROUP.

Dear Sir,

Thank you for accepting our manuscript for publication in BMC Cancer. The formatting changes you requested have been addressed as follows:

1. The full stop has been removed from the title.
2. There was no funding. Therefore, the Acknowledgement section remains unchanged.
3. The Figures have been renumbered 1-4 and resubmitted. Figure legends and references to the figures have also been changed accordingly.
4. Figure titles have been removed from the image file

Thank you for accepting our manuscript.

Yours sincerely

Aristotle Bamias, PhD, MRCP (UK)